Author’s response to reviews

Title: Teleradiology in general practice has a positive impact on fracture diagnostics and treatment: a retrospective observational study

Authors:

Jac. JWM Jacobs (jjwmjacobs@gmail.com)
Jan PAM Jacobs (j.p.a.m.jacobs@rug.nl)
Eric van Sonderen (E.L.P.van.Sonderen@umcg.nl)
Thys van der Molen (t.van.der.molen@med.umcg.nl)
Robbert Sanderman (R.Sanderman@umcg.nl)

Version: 2
Date: 31 July 2014

Author’s response to reviews:

Dear Mrs/Sir

Line numbering is completed.

The ethics and patients anonymity is described in main text.

It is an observational study: STROBE is used.

Data collection is described in the main text

As reported in the main text, no conflict of interest is reported.